\begin{table} %t2
\caption{\label{maintable}Radial velocities and space motions of our targets. Data on distance and proper motions are obtained from the literature.}
%\centering
$$
\begin{tabular}{llr@{ $\pm$ }lr@{ $\pm$ }rr@{ $\pm$ }rr@{ $\pm$ }lr@{ $\pm$ }lr@{ $\pm$ }lr@{ $\pm$ }lcl} 
\hline\hline\noalign{\smallskip}            
Object & SpT & \multicolumn{2}{c}{Distance}  & \multicolumn{2}{c}{$\mu_{\alpha}\cos{\delta}$} &  \multicolumn{2}{c}{$\mu_{\delta}$} & \multicolumn{2}{c}{RV} & \multicolumn{2}{c}{$U$} & \multicolumn{2}{c}{$V$} & \multicolumn{2}{c}{$W$} & Ref. & Comments\\
       &     & \multicolumn{2}{c}{(pc)}  & \multicolumn{2}{c}{(mas/yr)} &  \multicolumn{2}{c}{(mas/yr)} & \multicolumn{2}{c}{(km~s$^{-1}$)} & \multicolumn{2}{c}{(km~s$^{-1}$)} & \multicolumn{2}{c}{(km~s$^{-1}$)} & \multicolumn{2}{c}{(km~s$^{-1}$)} &   &  \\
\hline
2MASS J03140344+1603056 & L0.0 & 14.0 & 1.0 &  --241 & 18 &   --76 & 19 &  --8.0 & 1.1 &   16.4 & 1.4 &    5.4 & 1.4 &   --7.0 & 1.5 & 1 &     UMa moving group \\
2MASS J12212770+0257198 & L0.0 & 19.0 & 1.0 &  --115 & 30 &   --18 & 27 &  --9.0 & 1.4 &   --9.2 & 2.8 &   --2.6 & 2.5 &  --10.0 & 1.7 & 1 &           Young disk \\
2MASS J17312974+2721233 & L0.0 & 12.0 & 1.0 &   --82 & 15 &  --240 & 17 & --30.5 & 1.1 &   --5.8 & 1.4 &  --30.0 & 1.3 &  --14.3 & 1.0 & 1 & AB Dor moving group  \\
2MASS J22000201-3038327 & L0.0 & 35.0 & 2.0 &   210 & 48 &   --64 & 21 & --25.3 & 1.0 &  --39.7 & 6.7 &  --19.7 & 3.8 &   --1.6 & 5.2 & 1 & Hyades moving group  \\
2MASS J07464256+2000321 & L0.5 & 12.2 & 0.04 &  --374 &  1 &   --57 &  0 &  53.0 & 1.1 &  --55.3 & 1.0 &  --15.1 & 0.4 &   --1.0 & 0.4 & 1 &                      \\
2MASS J14122449+1633115 & L0.5 & 25.0 & 2.0 &    29 & 16 &   --80 & 30 &   5.0 & 1.1 &   10.0 & 2.8 &   --4.8 & 3.1 &    1.7 & 1.5 & 1 &           Young disk \\
2MASS J14413716-0945590 & L0.5 & 27.5 & 2.7 &  --198 &  3 &   --16 &  4 & --28.3 & 1.1 &  --34.7 & 1.8 &  --12.5 & 2.0 &  --10.4 & 1.3 & 1 & Hyades moving group  \\
2MASS J23515044-2537367 & L0.5 & 18.0 & 2.0 &   387 & 21 &   163 &  9 &  --3.0 & 1.1 &  --35.7 & 4.4 &   --2.1 & 1.1 &   --4.1 & 1.4 & 1 &           Young disk \\
2MASS J02355993-2331205 & L1.0 & 21.3 & 0.5 &    84 &  1 &    13 &  0 &  15.3 & 1.1 &  --11.7 & 0.4 &   --7.7 & 0.3 &  --10.5 & 1.1 & 2 &           Young disk \\
2MASS J06023045+3910592 & L1.0 & 10.6 & 0.3 &   146 & 10 &  --501 & 10 &   7.6 & 1.1 &  --11.4 & 1.1 &  --24.4 & 0.9 &   --4.4 & 0.6 & 1 & Pleiades moving group \\
2MASS J10224821+5825453 & L1.0 & 20.0 & 1.0 &  --360 & 99 &  --780 & 90 &  19.7 & 1.1 &  --40.0 & 8.1 &  --69.7 & 9.3 &   23.7 & 6.2 & 1 &                      \\
2MASS J10452400-0149576 & L1.0 & 17.0 & 1.0 &  --495 & 18 &    12 & 12 &   7.0 & 1.1 &  --35.7 & 2.5 &  --14.2 & 1.3 &  --12.9 & 1.6 & 1 & Hyades moving group \\
2MASS J10484281+0111580 & L1.0 & 15.0 & 1.0 &  --442 & 13 &  --209 & 12 &  24.0 & 1.1 &  --24.9 & 1.6 &  --34.0 & 1.7 &   --2.2 & 1.8 & 1 &                      \\
2MASS J13004255+1912354 & L1.0 & 14.0 & 1.0 &  --820 & 18 & --1244 & 19 & --17.8 & 1.0 &   --1.6 & 1.3 &  --96.8 & 7.4 &  --27.0 & 1.3 & 1 &                      \\
2MASS J13595510-4034582 & L1.0 & 21.0 & 1.0 &    38 & 11 &  --485 & 15 &  49.8 & 1.0 &   39.3 & 1.1 &  --50.5 & 1.5 &  --27.0 & 2.6 & 1 &                      \\
2MASS J14392836+1929149 & L1.0 & 14.4 & 0.1 & --1229 &  1 &   407 &  2 & --27.2 & 1.1 &  --82.6 & 0.7 &  --39.2 & 0.3 &   14.1 & 1.1 & 1 &                      \\
2MASS J15551573-0956055 & L1.0 & 13.0 & 1.0 &   950 & 15 &  --767 & 15 &  14.5 & 1.1 &   52.2 & 3.4 &    2.3 & 1.0 &  --56.1 & 5.2 & 1 &                      \\
2MASS J11455714+2317297 & L1.5 & 44.0 & 3.0 &   155 & 16 &   --56 &  6 &   3.3 & 1.0 &   32.9 & 3.9 &    2.4 & 1.9 &   10.3 & 1.4 & 1 &                      \\
2MASS J13340623+1940351 & L1.5 & 46.0 & 3.0 &   --58 & 12 &    98 & 16 &   2.0 & 4.1 &  --22.0 & 3.5 &    9.3 & 3.4 &    6.9 & 4.2 & 1 &                      \\
2MASS J16452211-1319516 & L1.5 & 12.0 & 1.0 &  --364 & 18 &  --804 & 16 &  26.4 & 1.0 &   32.6 & 1.2 &  --46.4 & 4.3 &   --0.3 & 1.3 & 1 &                      \\
2MASS J18071593+5015316 & L1.5 & 14.0 & 1.0 &    35 & 19 &  --126 & 14 &  --2.0 & 3.2 &    7.6 & 1.3 &   --1.7 & 2.9 &   --4.3 & 1.9 & 1 &           Young disk \\
2MASS J20575409-0252302 & L1.5 & 16.0 & 1.0 &    10 & 20 &   --80 & 20 & --25.0 & 3.2 &  --12.9 & 2.3 &  --20.3 & 2.4 &    9.0 & 2.1 & 1 &           Young disk \\
2MASS J08283419-1309198 & L2.0 & 11.6 & 1.4 &  --592 &  6 &    14 &  7 &  26.2 & 1.0 &  --33.1 & 2.5 &  --16.6 & 1.1 &  --19.4 & 3.3 & 3 &           Young disk \\
2MASS J09211410-2104446 & L2.0 & 12.0 & 1.0 &   244 & 16 &  --908 & 17 &  80.0 & 1.1 &   18.1 & 3.8 &  --94.4 & 2.3 &    5.1 & 2.2 & 1 &                      \\
2MASS J11553952-3727350 & L2.0 & 13.0 & 1.0 &    50 & 12 &  --767 & 15 &  45.0 & 1.1 &   34.4 & 1.8 &  --50.6 & 1.5 &  --22.9 & 3.4 & 1 &                      \\
2MASS J13054019-2541059 & L2.0 & 18.7 & 0.7 &  --285 &  1 &    11 &  1 &   5.0 & 2.2 &  --18.7 & 1.4 &  --17.0 & 1.6 &    5.2 & 1.4 & 1 &           Young disk \\
2MASS J05233822-1403022 & L2.5 & 13.0 & 1.0 &    90 & 17 &   166 & 17 &  11.3 & 1.0 &  --15.7 & 1.2 &   --2.6 & 1.1 &    3.4 & 1.3 & 1 &           Young disk \\
2MASS J10292165+1626526 & L2.5 & 23.0 & 2.0 &   359 & 15 &  --348 & 17 & --28.2 & 1.4 &   58.5 & 4.5 &  --14.3 & 2.9 &  --11.8 & 1.9 & 1 &                      \\
2MASS J10473109-1815574 & L2.5 & 22.0 & 2.0 &  --347 & 17 &    54 & 14 &   6.0 & 1.1 &  --34.0 & 3.6 &  --11.7 & 1.5 &   --9.3 & 1.9 & 1 & Hyades moving group \\
2MASS J09130320+1841501 & L3.0 & 46.0 & 4.0 &    32 & 16 &  --187 & 17 &  --1.0 & 2.2 &   17.1 & 3.5 &  --37.1 & 5.0 &   --6.4 & 3.2 & 1 &                      \\
2MASS J12035812+0015500 & L3.0 & 19.0 & 2.0 & --1209 & 18 &  --261 & 15 &  --1.5 & 2.2 &  --83.7 & 9.2 &  --66.3 & 7.5 &  --31.9 & 3.9 & 1 &                      \\
2MASS J15065441+1321060 & L3.0 & 14.0 & 1.0 & --1087 & 13 &    14 & 11 &  --0.9 & 1.2 &  --41.4 & 3.2 &  --47.4 & 3.6 &   35.3 & 2.9 & 1 &                      \\
2MASS J16154416+3559005 & L3.0 & 24.0 & 2.0 &   --17 & 12 &  --512 & 15 & --21.0 & 1.2 &   42.1 & 4.6 &  --42.5 & 3.0 &  --16.1 & 1.3 & 1 &                      \\
2MASS J21041491-1037369 & L3.0 & 17.0 & 2.0 &   614 & 16 &  --281 & 15 & --20.5 & 2.2 &  --37.9 & 3.5 &  --28.9 & 2.7 &  --33.4 & 5.7 & 1 &                      \\
2MASS J00361617+1821104 & L3.5 &  8.8 & 0.1 &   899 &  1 &   120 &  2 &  21.0 & 1.8 &  --41.0 & 0.7 &   --3.2 & 1.2 &  --13.3 & 1.3 & 1 &           Young disk \\
2MASS J07003664+3157266 & L3.5 & 12.2 & 0.3 &   130 &  1 &  --546 &  1 & --43.5 & 2.2 &   43.0 & 2.2 &  --28.6 & 0.8 &  --16.8 & 0.6 & 1 &                      \\
2MASS J17054834-0516462 & L4.0 & 11.0 & 1.0 &   129 & 14 &  --103 & 15 &  12.2 & 1.1 &   14.2 & 1.1 &    2.8 & 0.8 &   --3.7 & 1.1 & 1 &     UMa moving group \\
2MASS J22244381-0158521 & L4.5 & 11.4 & 0.1 &   457 &  2 &  --871 &  1 & --39.0 & 1.1 &  --10.4 & 0.4 &  --64.9 & 0.8 &   --5.0 & 0.9 & 1 &                      \\
2MASS J00043484-4044058 & L5.0 & 13.0 & 0.7 &   643 &  2 & --1494 &  1 &  30.0 & 1.4 &   11.4 & 0.5 & --102.9 & 5.4 &  --15.3 & 1.6 & 4 &                      \\
2MASS J08251968+2115521 & L5.0 & 10.7 & 0.1 &  --510 &  2 &  --288 &  2 &  20.0 & 3.2 &  --27.5 & 2.6 &  --16.9 & 1.1 &  --15.5 & 1.6 & 1 &           Young disk \\
2MASS J08354256-0819237 & L5.0 &  7.0 & 1.0 &  --530 & 30 &   300 & 30 &  26.5 & 2.2 &  --31.2 & 1.6 &  --11.8 & 1.9 &   --0.5 & 1.2 & 1 & Hyades moving group \\
2MASS J15074769-1627386 & L5.0 &  7.3 & 0.03 &  --161 &  2 &  --888 &  0 & --40.4 & 1.8 &  --26.4 & 1.5 &  --16.7 & 0.4 &  --40.5 & 1.1 & 1 &                      \\
2MASS J02550357-4700509 & L8.0 &  5.0 & 0.1 &   999 &  2 &  --565 &  3 &  25.0 & 4.1 &   --5.4 & 0.4 &  --35.7 & 2.2 &   --7.3 & 3.6 & 5 &                     \\                                  
\hline
\end{tabular}
$$
{\bf Notes.} References -- Distances and proper motion: (1) \citet[][and references therein]{Faherty09}, 
(2) HIPPARCOS catalogue  \citep{Perryman97}, (3) \citet{Lodieu05}, (4) \citet{Henry06}, (5) \citet{Costa06}.
\end{table}